

Ethnobiology and Conservation, 12:06 (16 February 2023) doi:10.15451/ec2023-02-12.06-1-15 ISSN 2238-4782 ethnobioconservation.com

# Social perceptions of ecosystem services delivered by coastal wetlands: their value and the threats they face in northwestern Mexico

Aimée Cervantes-Escobar<sup>1</sup>, Arturo Ruiz-Luna<sup>1\*</sup> and César Alejandro Berlanga-Robles<sup>1</sup>

## ABSTRACT

Wetlands are recognized for their socio-environmental value and capacity to provide ecosystem services (ES) and are vulnerable to diverse threats, including those derived from climate change (CC). However, changes in ES delivery may not be recognized by ES users. To determine the extent to which coastal communities in northwestern Mexico are aware of the presence and importance of coastal wetlands, participatory workshops that followed the Metaplan methodology were held in four communities. The effects produced by extreme rainfall events (considered manifestations of CC) on wetlands and their ES were also analyzed. Four coastal wetland types (estuaries, saltmarshes, mangroves, and lagoons) were perceived to be the most important ecosystems, while poor fishing practices, mangrove deforestation, and pollution were identified as their main threats. Climate change, land use changes, and water mismanagement were also perceived as wetland threats. There were a few differences among communities that were mostly related to the number of ES and the ES categories identified. Nonetheless, saltmarshes and mangroves were identified as priority wetlands in the communities included in this study, and their ES were recognized. Most of the participants in all communities agreed on the main threats facing wetlands and their ES, particularly those related to CC and those that affect their livelihoods. Despite this awareness, regional coastal wetlands continue to decline. As such, participatory methodologies are needed to firmly establish the importance of wetlands within communities. In doing so, alternative actions based on traditional knowledge can be integrated into management actions, and novel solutions may be developed at the community level that can be scaled regionally.

Keywords: Climate change, Ecosystem services, Extreme rainfall events, Social perception, Wetlands.

 $<sup>1 \ {\</sup>rm Centro} \ {\rm de \ Investigación \ en \ Alimentación \ y \ Desarrollo \ A.C. \ Av. \ Sábalo \ {\rm Cerritos \ S/N, \ 82112. \ Mazatlán, \ Sinaloa, \ México. \ No \ Sabalo \ {\rm Mexico. \ Sabalo \ \ Sabalo \ Sabalo \ Sabalo \ Sabalo \ Sabalo \ Sabalo$ 

<sup>\*</sup> Corresponding author 🖾. E-mail address: ACE (acerva.a@gmail.com), ARL (arluna@ciad.mx), CABR (cesar@ciad.mx).

#### SIGNIFICANCE STATEMENT

With the aim to understand the extent to which inhabitants of coastal communities recognize the presence and importance of local wetlands and their environmental services, participatory workshops were held in four localities in northwestern Mexico. We used the Metaplan methodology to identify similarities and differences among communities in the perceptions of inhabitants regarding local wetlands. Almost all wetlands included in this study were recognized by the participants. Most participants were aware of the main threats to wetlands, particularly those associated with climate change or those that affect livelihoods, and the services these ecosystems provide. As regional coastal wetlands continue to decline, it is necessary to use participatory methodologies to establish their importance within communities. In doing so, alternative actions based on traditional knowledge can be integrated into management actions, and novel solutions may be developed at the community level that can be scaled regionally.

#### INTRODUCTION

Wetlands are of great ecological, economic, and social importance. These ecosystems are found throughout the world and are characterized by having substrata that are either temporarily or permanently saturated with water. In addition, wetlands are covered, at least periodically, by aquatic or hydrophilic vegetation, which is a remarkable feature of these ecosystems (Mitsch and Gosselink 2015). Wetlands provide critical foraging, nesting, and nursery habitats for resident or migratory species, such as waterbirds and shorebirds, fishes, frogs, and crustaceans, and play important roles in maintaining terrestrial and aquatic ecosystems (Vernon et al. 2019). Due to their regional importance or geographic expanses, some wetlands have been designated as important national or international sites (e.g., Ramsar sites) that are protected by legal instruments to ensure they are properly conserved and sustainably used (e.g., North American Wetlands Conservation Act; CONABIO-Mexico).

The different wetland types found in inland and marine environments provide diverse ecosystem services (ES), which have been defined as the ecological characteristics, functions, or processes that directly or indirectly contribute to human wellbeing (Costanza et al. 2017). Ecosystem services may be broadly characterized as provisioning, regulating, supporting, and cultural services that promote human safety and welfare (Camacho-Valdez and Ruiz-Luna 2012). Wetland-derived ES include supplying animal and plant proteins, purifying water and ensuring its availability, regulating climate and hydrological regimes, and protecting against erosion (Millennium Ecosystem Assessment 2005). Some wetlands also show notably high capacities to sequester carbon. In fact, 20-30% of all carbon stored in soils worldwide is found in wetland-associated areas (Mitsch 2016). In this context, coastal wetlands supply important ES by regulating climate, sequestrating carbon (i.e., blue carbon), protecting against storms and hurricanes, regulating coastal erosion, and providing refuge to various species of commercial interest (Camacho et al. 2014; Rojas et al. 2017).

Despite their importance, the loss or degradation of wetlands is increasing worldwide, and the remaining wetlands are directly or indirectly at risk due to human activities (TEEB 2013). From the beginning of the 20th century to the present, 64-71% of all wetland cover has been lost (Davison 2014). Mitsch and Gosselink (2015) highlighted wetland losses in the United States (53%), Australia (50%), China (60%), and New Zealand (90%) while reporting relatively minor losses in boreal countries and extreme losses (> 90%) in parts of Europe. These losses have been caused by various anthropogenic drivers such as the development of hydraulic infrastructure (e.g., channels, dams, and dikes), changes in land use, pollution, urban sprawl, and the introduction of exotic species. Overall, the main cause of inland wetland degradation is water drainage for agricultural development, whereas changes in land cover and land use are the main causes of degradation in coastal wetlands (Millennium Ecosystem Assessment, 2005; Clarkson et al. 2013). In some tropical and subtropical coastal regions, the expansion of shrimp farming has also been identified as an important driver of wetland loss (Ottinger et al. 2016).

In addition to the aforementioned anthropogenic factors, global change also threatens the existence and resilience of wetlands. Global change refers to a set of natural environmental changes that are heightened by anthropogenic activities, including the increase in atmospheric  $CO_2$ , climate change, and ongoing changes in land use/land cover (Sage 2020). These changes not only have environmental consequences, such as those derived from increasing temperatures or air and water pollution and those resulting from extreme weather events (e.g., droughts, storms, tornadoes, hurricanes, and extreme precipitation), but also generate disease and reduce productivity, which have negative repercussions for human wellbeing and the economy (Duarte 2006). Global change can also alter hydrological and geomorphological features of coastal wetlands and their community structure on local and regional

scales, which can affect natural selection, extinction rates, biodiversity, energy flows, productivity, and nutrient cycles in these and other ecosystems (Flores et al. 2010). As such, Carpenter et al. (1992) predicted that the distributions and expanses of coastal wetlands would be modified by climate change.

The Stern Report (Stern and Stern 2007) indicates that climate change tends to increase species extinction rates, reduce water availability, intensify the disappearance of mountain glaciers, and promote a rise in mean sea level, all of which threaten the existence of wetlands. These threats culminate in a greater risk of wetland loss in developing countries when compared to the risk present in developed countries due to the interplay between climate change and the pre-existing social and economic vulnerabilities of each country (Emérit 2008).

In Mexico, there is no precise estimate of the expanse of wetlands within the country. In fact, estimates are notably different among sources, such as those from Olmsted (1993; 33,000  $km^2$ ), Carrera and de la Fuente (2003; 45,000  $km^2$ ), and the Ministry of the Environment (SEMARNAT 2012; 128,000  $km^2$ ). The estimate from the Ministry of the Environment indicates that wetlands cover approximately 6.5% of the national territory. Discrepancies among these and other estimated values have been well documented in some coastal wetlands (e.g., mangroves; Ruiz-Luna et al. 2008) and have been attributed to the absence of a standardized classification system and methodological framework to evaluate and monitor the expanses of these ecosystems. Recently, important initiatives have been put forth to address this issue, and a national standardized methodology for wetland evaluations has been developed, which was led by efforts by the National Water Commission (CONAGUA 2017). However, the results of this initiative have not yet been published. Regardless of the differences among estimates, a clear decline in wetland cover and functionality is apparent. A poor understanding of the roles these ecosystems play in ensuring human wellbeing by providing and maintaining ES (Maltby and Acreman 2011) has led to the development of insufficient conservation measures at both regional and local scales.

Given the existing inadequacy of management strategies in light of the direct mid- and long-term environmental effects of climate change and the ever increasing risk of wetland deterioration, the present study aimed to identify the perceptions of the inhabitants of coastal communities in northwestern Mexico regarding the importance of their local wetlands. This study also includes an analysis of the ES provided by wetlands and how these ES may have been affected by an increase in extreme rainfall events (ERE). The results of this study serve to improve and support management strategies by elucidating the knowledge held by local communities and encouraging their active participation in future conservation efforts.

# MATERIAL AND METHODS

#### Study area

This study was conducted in northwestern Mexico in four coastal communities in the states of Sinaloa and Baja California Sur (BCS) that border the Gulf of California: Cristo Rey (CR), Isla del Bosque (IB), Las Arenitas (LA), and Loreto (LR), (Figure 1). Both CR and IB are located in southern Sinaloa in the Escuinapa municipality, while LA is located in the Culiacan municipality of Sinaloa, and LR is located in the municipality of the same name in BCS. Each locality is associated with wetlands or wetland complexes that have been designated as Ramsar sites (RS). The wetlands of CR and IB belong to Marismas Nacionales (RS No. 732), and the wetlands of LA belong to the Altata-Ensenada de Pabellones Lagoon System (RS No. 1760). The wetlands of LR belong to the Bahía de Loreto National Park in BCS (RS No. 1358). The study sites are also found within areas classified as priority hydrological, marine, and terrestrial regions by the Commission for the Knowledge and Use of Biodiversity of Mexico (CONABIO 2022).

The population of the Escuinapa municipality totals 61,644 inhabitants, 44.1% of which live in poverty (Coneval 2021). The main economic activities in this coastal area are agriculture (mainly fruits), fishing, and aquaculture. A total of 1,064,328 inhabitants reside in the Culiacan municipality (INEGI 2020), 23% of which live in poverty (Coneval 2021). The economy of this municipality is also based on agriculture, aquaculture, and fishing, although livestock are also raised. The Loreto municipality had a total population of 18,052 inhabitants in 2020, 34% of which lived in poverty, 20.4% of which were employed in commercial activities, 18.5% of which were employed in temporary or seasonal hospitality services, and 12.9% of which were employed in primary activities like agriculture, animal husbandry, forestry, fishing, or hunting (Data Mexico 2021).

## Workshops and Data Collection

Participatory workshops were held in each study community with local stakeholders and wetland users. The workshops followed the Metaplan qualitative methodology to structure, manage, and facilitate group communications (Habershon 1993). This methodology allows for problems and possible solutions to be identified with the opinions of participants expressed on small cards (13 x 18 cm). Different co-



Figura 1. Study area and workshop locations. The workshops were held in four study sites located in four coastal communities in the states of Sinaloa and Baja California Sur (BCS) that border the Gulf of California: Cristo Rey (CR), Isla del Bosque (IB), Las Arenitas (LA), and Loreto (LR).

lored cards were used for each phase of the workshop, which allowed the participants to express and share their own opinions after a phase had concluded by posting their cards on a wall or blackboard. This methodology reduced the number of redundant items, which ensured that the ideas of all participants were properly organized and accounted for at the end of the workshop.

The workshops began by introducing the team of facilitators. This was followed by a brief introduction to the different types of wetlands and ES, which included graphical representations and explanations to establish a common language among participants. The importance of wetlands as ES providers was emphasized, and their classifications according to the Millennium Ecosystem Assessment were explored (MEA 2005). Subsequently, a series of activities were conducted that aimed to determine what the participants knew about the wetlands in their region, their importance, and the possible threats to these ecosystems. After which, the ES offered by wetlands, their local importance, and the possible changes induced by torrential rains or ERE were established.

Based on the Metaplan method, each participant was asked to confirm the presence or absence in the region of ten common wetland types. The participants were also allowed to include additional wetland types from a list provided by the facilitators. Subsequently, the participants were asked to reflect and choose 3 to 5 wetland types whose loss or deterioration would affect them personally and to identify the wetland types that they considered to be important for their community. The first phase of the workshop concluded with the identification of possible threats to the continued existence of these ecosystems (e.g., deforestation, contamination, drying out, or channeling). The threats were collectively listed to record their number by priority wetland type and community and to minimize the effect of sample size in the analysis.

In the second stage of the workshop, the participants focused on identifying the specific ES provided to each community by each priority wetland type. First, the participants identified the ES offered by each priority wetland type (e.g., fishing, the provision of materials, climate regulation, and opportunities for recreation). These were then grouped into one of the following categories proposed by the Millennium Ecosystem Assessment: regulating and supporting (R; joined here in one class for operational purposes) and provisioning (P) and/or cultural (C). Subsequently, the participants rated the importance of the ES based on a four-level ordinal scale: 1) not important, 2) not very important, 3) important, and 4) very important. Finally, the perceptions of the participants regarding the effects of ERE on the capacity of wetlands to supply ES were recorded using the following scale: 1) the capacity to supply ES is lost, 2) the capacity to supply ES decreases, 3) the capacity to supply ES remains the same, and 4) the capacity to supply ES increases.

The workshops were concluded by integrating the information that had been generated with the direct or indirect impacts of ERE on each community. Four types of impacts were identified: floods (Fl), pollution (Po), siltation (St), and water turbidity (Tu). In addition, seven types of change were identified: diminished fishing activities (DF), loss of crops (LC), shortage of food and water (SW), effects on tourism (ET), house damage (HD), health problems (HA), and disruptions to the supply of electricity (E). The participants were asked to classify each according to their experience as either primary (1) or secondary (0).

# Data analysis

As part of a wetland prioritization exercise, the votes that each wetland received in each community were ranked in a two-way contingency table to evaluate the homogeneity (independence) hypothesis with an extended Fisher exact test (West and Hankin 2008). A post-hoc pairwise Fisher exact test was then conducted (MacDonald and Garner 2000; McDonald 2014). After the prioritization analysis, information was collected for only the priority wetlands identified by each community. However, discrepancies prevented the response and predictor variables from being ordered in balanced experimental designs, so the subsequent analyses of variance (ANOVAs) were conducted with random incomplete block designs.

The effect of community (factor) on the number of threats identified (response) was analyzed via an ANOVA for a Poisson response (analysis of deviance with the X2 statistic) considering the block effect of wetland type and, inversely, the effect of wetland type and the block effect of community. These analyses were followed by a post-hoc analysis of paired comparisons of the least squares means (lsmeans) with a Tukey adjustment (Agresti 2007; Mangiafico 2015). Similarly, the number of identified ES were analyzed including the factor and block effects of the different ES categories.

The perceptions of the importance of ES and the capacity of wetlands to continue offering them after an ERE were analyzed via ANOVAs for ordinal responses and the factor and block effects of community, wetlands, and wetland categories (provisioning, cultural, and regulating /supporting) in the first case and those of community and wetlands in the second. In addition, responses concerning the importance of the impacts of ERE and the changes they induce were analyzed with ANOVAs for binomial responses with the factor and block effects of the community and the type of impact or change.

# RESULTS

A total of 53 people attended the different workshops. The largest workshop hosted 20 participants while the smallest hosted seven people. The participation of women in LR and LA was minimal (1 and 0, respectively), while the number of women was 5 and 12 in CR and IB, respectively. The participants identified 4-8 of the 10 wetlands on the list provided by the moderators. In addition, the participants of the LR workshop also identified oases, which are only present in their region, and designated them as priority wetlands. Overall, five priority wetlands were identified in CR, and four priority wetlands were identified in the other three communities. Lakes and reedbeds were not recognized by participants, although they appeared on the wetland list used in the workshops, while estuaries and mangroves were identified as priority wetlands (Table 1).

In the three communities of Sinaloa, saltmarshes and shrimp ponds were considered priority wetlands, but despite these similarities, the differences were greater (p < 0.001 with 10,000 iterations) and were mainly due to the prioritization of saltmarshes with respect to other wetland types (p < 0.05) except shrimp ponds (p > 0.1). In addition, differences were present with regard to the prioritization of wetland types between IB and LA (p < 0.01) and LR (p < 0.001) and between LR and CR (p < 0.01).

In the communities of southern Sinaloa (CR and IB), threats to wetlands include poor fishing practices, mangrove deforestation, and diverse pollution sources. In LA, the fishing practice of *enyerbado*, which

# Ethnobiol Conserv 12:06

**Tabela 1.** Wetlands identified and prioritized in the four coastal communities of northwestern Mexico included in this study: Cristo Rey (CR), Isla del Bosque (IB), Las Arenitas (LA), and Loreto (LR). Symbols: (+) wetland identified within the municipal limits, (-) wetland not identified within the municipal limits, and (p) wetland designated as priority. The number of votes received in the prioritization exercise appears in parenthesis. The colors indicate the prioritization level: from gray for wetlands identified but not prioritized to green for wetlands with the most votes in each community. Brown and pink represent intermediate values.

Wetland type	Description	Community			
		$\mathbf{CR}$	IB	$\mathbf{L}\mathbf{A}$	$\mathbf{LR}$
Estuaries	Coastal, brackish water bodies highly influen- ced by tides and river discharge	+p(3)	+p(1)	+p(6)	+p(7)
Saltmarshes	Coastal wetlands with unconsolidated substrata dominated by herbaceous species	+p(6)	+p(16)	+p(1)	+
Mangroves	Coastal wetlands with unconsolidated subs- trata dominated by shrubs or trees	+p(2)	+p(5)	+p(2)	+p(7)
Lagoons	Coastal water bodies that are usually shallow and separated from the sea by narrow land- forms	+	+	-	+
Beaches	Landforms next to water bodies (sea and lago- ons in the present study) influenced by tides with unconsolidated soils	+p(1)	+	-	+p(2)
Reedbed	Vegetation type including diverse plants adapted to growing in wet conditions	-	-	-	-
Oases*	Areas in arid regions where the water table is near or above the surface creating springs or wells	-	-	-	+p(6)
Lakes	Inland water bodies filled mostly with freshwa- ter	-	-	-	-
Rivers and creeks	Channeled water streams	+	+	-	+
Shrimp ponds	Artificial wetlands for controlled shrimp pro- duction	+p(1)	+p(2)	+p(1)	+
Dams	Artificial barriers to contain flowing water and create reservoirs	+	+	-	+
Identified wetlands		8	8	4	9

Note: \*Wetland type added by the LR participants to the original list of 10 wetland types provided by workshop facilitators.

employs poison, and the construction of shrimp ponds were identified as the main wetland threats. In contrast, the threats identified in LR were diverse and included global change and climate change and their effects (e.g., hurricanes and droughts), land use change for tourism purposes, real estate developments, inefficient water management practices, and pollution.

The number of threats varied from 0 for shrimp ponds in CR to 18 for estuaries in LR (Figure 2). Differences among communities regarding the number of identified threats (p < 0.001) were due to the number of threats identified in LR (lsmean = 14.4) with respect to the Sinaloa communities (p < 0.005). Four times the number of threats were identified in LR compared to those identified in CR (lsmean = 3.4) and LA (lsmean = 3.6). In addition, the number of threats in LR was double that of IB (lsmean = 6.4).

The differences among wetlands based on the number of threats were not significant (p > 0.1) due to the block effect of community. In general, more threats were identified in LR for all wetland types, while the number of threats identified for each priority wetland type in each of the three communities of Sinaloa were similar. In addition, the value of the Spearman correlation between the number of votes received in the prioritization exercise and the number of threats identified per community and wetland type was 0.62 (p < 0.01).

The relationship between ES and wetland type varied among sites. In all, 17–39 ES were identified among all wetland types. In Sinaloa, a greater number of ES were associated with saltmarshes, estuaries, and mangroves. The participants of the CR and LA workshops identified 17 ES for each site that were mainly associated with mangroves (35%) and saltmarshes (41%), respectively. In IB, ES were associated with estuaries and mangroves (39% for each wetland type). In addition, 17 ES were identified in the LA workshop, with around 41% of them being linked to saltmarshes. As with the number of threats, a markedly higher number of ES (39) were identified in the LR workshop when compared to those identified in the other workshops. These ES were almost evenly distributed among the priority wetlands identified in LR (Figure 3).

In all four workshops, provisioning services were identified for each priority wetland type with the exception of beaches in CR. These ES represent 25-100% of the services identified in this study and were mainly related to commercial and subsistence fishing (Figure 3). In addition, regulating/supporting and cultural services were recognized as being delivered by the priority wetlands identified in the four workshops. Other ES included protection against storms and hurricanes: oxygen production: carbon sequestration; flood regulation; breeding, nesting, and refuge sites for various species of ecological and economic importance; areas for recreation (e.g., swimming, fishing, and relaxing) and research activities; tourism; scenic beauty; and cultural identity. The ES offered by shrimp ponds were recognized in the three workshops in Sinaloa. In IB, cultural services were also identified for this wetland type, although these were more linked to human well-being and health. Lastly, in LR, the workshop participants acknowledged receiving the benefits from the four priority wetland types that belong to the aforementioned categories.

Intra-community differences in the number of identified ES were detected with the wetland (p < 0.01) and ES category (p < 0.05) block effects. These were especially evident in the comparison of the wetland evaluation between CR and LR (p < 0.05) and in the differences between the category evaluations between the LR and Sinaloa communities (p < 0.05). Likewise, discrepancies were found among the participants of the four workshops in the number of services identified for each wetland type when they were

grouped in blocks by ES category (p < 0.01). These were particularly evident between shrimp ponds and estuaries (p < 0.05), shrimp ponds and mangroves (p < 0.01), and mangroves and oases (p < 0.05) and to a lesser extent, between mangroves and beaches (p < 0.1).

About 51% of all participants said that the ES of priority wetlands are very important, while 32% of all participants indicated that the priority wetlands provide important ES. In CR, only 18% of participants considered that the cultural services provided by estuaries and the provisioning services offered by mangroves were not important, although 82% of participants considered that the provisioning and regulation ES provided by saltmarshes were very important. In IB, 71% and 75% of the participants considered the mangrove provisioning and regulation ES to be very important, respectively, whereas 40% and 32% of participants rated the cultural services of shrimp ponds and estuaries to be not very important, respectively. In LA, the provisioning ES of saltmarshes were not very important while mangrove provisioning and regulation ES were ranked very important. In LR, 71% of the participants rated the provisioning ES of oases as very important, while 13% and 17% of participants ranked the same services provided by estuaries and mangroves to be not very important (Figure 3).

Both the wetland type and ES category were considered in each community to assess the importance of ES (p < 0.001), and differences were found among communities ( $p \ge 0.05$ ) with the exception of the comparison between CR and LR (p > 0.1). In addition, differences were detected in how wetlands were valued when considering the effect of the category (p < 0.001). This was mainly due to saltmarshes and oases being valued higher than beaches and shrimp ponds (p < 0.05). Likewise, ES were valued differently among communities (p < 0.001) due to the greater importance that was assigned to regulating and supporting services (p < 0.001).

When evaluating the capacity of the wetlands to continue offering ES after an ERE, 78% of all workshop participants believed that this capacity would change, and 58% believed that it would decrease, although differences were present among communities (p < 0.01; Figure 4).

The pairwise comparisons revealed agreement between IB and LR (p > 0.1). The participants in these communities tended to think that the capacity of priority wetlands to provide ES would be maintained or even increase after an ERE. In CR and in LA in particular, participants tended to believe that this capacity would decrease or be lost. These differences (p < 0.001) were mainly due to participants believing that shrimp ponds had a low capacity to provide ES when compared to the capacity of natural priority we-



**Figura 2.** Number of threats identified for the priority wetlands of four coastal communities in northwestern Mexico: Cristo Rey (CR), Isla del Bosque (IB), Las Arenitas (LA), and Loreto (LR). Wetland type: estuaries (Est), marshes (Mrs), mangroves (Mng), beaches (Bch), oases (Oas), and shrimp ponds (Shr).

tlands (p < 0.001). In addition, these differences were attributed to the high perceived capacity of oases to provide services, which contrasted with the perceived capacities of estuaries, mangroves, and beaches to do the same ( $p \le 0.01$ ), although this was not true for saltmarshes (p > 0.1).

Finally, considering the impacts and changes induced by an ERE that were identified by the communities, all of the impacts and 80% of the changes were defined as primary in IB. In LR and LA, 75% of the impacts, including pollution and siltation, were identified as primary. In LA, only two changes were considered to be primary, whereas only a food and water shortage was considered to be a primary change in LR (Figure 5). In CR, flooding was the only impact considered to be primary, but five of seven changes were also considered to be primary. This community was the only one to consider health problems and a disruption to the supply of electricity to be primary changes.

Given that the assessment of impacts and changes was collectively conducted, these variables were analyzed with a completely randomized block design that did not include health problems or disruptions to the supply of electricity in the ANOVA, as these were only added by the CR participants. In the analysis of the importance of the impacts among communities (factor) grouped by the type of impact (blocks) and the differences among impacts grouped by community, no differences were identified (p < 0.1). However, when analyzing changes, differences between communities and the type of change were detected (p < 0.01). Due to the presence of the adjusted probabilities of 0 and 1 in the logistic model, the post-hoc analyses were not able to identify significant differences.

## DISCUSSION

This study aimed to determine if community members in northwestern Mexico perceive threats to coastal wetlands that are due to human activities or the effects of global change, including the loss of ecosystem services, beyond those caused by heavy rainfall such as flooding or damage to infrastructure. This study fits within the framework of a broader project that aimed to determine the extent to which coastal communities in northwestern Mexico are vulnerable to the effects of ERE.

To this end, we utilized participatory workshops to collect data from each community. Participatory workshops were considered to be the most suitable format to encourage participants to actively and fre-

#### Ethnobiol Conserv 12:06



**Figura 3.** Number and importance of ecosystem services (ES) provided by wetlands in coastal communities in northwestern Mexico: Cristo Rey (CR), Isla del Bosque (IB), Las Arenitas (LA), and Loreto (LR). Wetland type: estuaries (Est), marshes (Mrs), mangroves (Mng), beaches (Bch), oases (Oas), shrimp ponds (Shr). The percentages reflect the number of ES (SN) with respect to the total SN identified by each community. Percentages are shown based on wetland type and ES category and their combination (i.e., importance). The ES categories of provisioning (P), regulating and supporting (R), and cultural (C) are given for each wetland type in each community.

ely express their opinions. However, ensuring that the workshops were properly designed to obtain sufficient information on a variety of topics needed for the subsequent quantitative analysis was challenging. The Metaplan methodology provided the necessary solutions to the challenges associated with designing a productive and participatory workshop. During the workshops, participants utilized cards to help structure their ideas and rank priorities. In the communities of southern Sinaloa (IB and CR), participants recognized many of the wetland types that had been previously identified by the workshop facilitators. Some artificial wetlands were also recognized (Berlanga-Robles et al. 2008; Camacho et al. 2013), whereas two continental wetland types were dismissed, and one wetland type was identified that had not been previously considered. This serves as indirect

evidence of the knowledge that workshop participants hold of their regional wetlands.

Between 15 to 20 participants (both men and women) with diverse occupations were expected for each workshop. However, one of the difficulties we encountered was to ensure a sufficient number of attendees for each workshop from diverse backgrounds, including women participants. This was especially important in communities in which vocations exist for specific economic activities. For example, fishing is an important vocation in LA, and there were no other types of users of the wetland or female assistants in this community. Ensuring that the participants of each workshop were representative of their community was important, as gender and occupational inequalities influence results given that perceptions change according to whether or not a resource is exploited or

#### Ethnobiol Conserv 12:06



**Figura 4.** Capacity of coastal wetlands to continue offering ecosystem services (ES) after an extreme rainfall event (ERE) based in the perceptions of the inhabitants of four coastal communities in northwestern Mexico: Cristo Rey (CR), Isla del Bosque (IB), Las Arenitas (LA), and Loreto (LR). Wetland type: estuaries (Est), marshes (Mrs), mangroves (Mng), beaches (Bch), oases (Oas), and shrimp ponds (Shr). Capacity: Loss (L), Decrease (D), No Change (NC), and Increase (I). The percentages in the capacity column reflect the number of individual votes received by each wetland with respect to the total number of votes of each community.

used by individuals or a sector. Carvajal (1994) and Marín-Muñiz et al. (2016) arrived at these conclusions. These authors studied perceptions of changes in land use and the use of wetland resources in Costa Rica and Veracruz (Mexico), respectively. Thus, the benefits derived from wetlands are not seen in the same way by all people. It is therefore recommended that future studies consider differences in gender, livelihoods, and even age to minimize bias.

In this study, the demographics of the workshop participants did not result in notable bias in terms of the number of wetlands or ES they identified. However, there were significant differences between priority wetlands by location, which did not always correspond to the number of ES that each wetland type provided according to the perceptions of the parti-

#### Ethnobiol Conserv 12:06



Figura 5. Importance of the impacts (A) and alterations (B) caused by an extreme rainfall event (ERE) in the four coastal communities in northwestern Mexico: Cristo Rey (CR), Isla del Bosque (IB), Las Arenitas (LA), and Loreto (LR). Importance: secondary (0) and primary (1). Impacts: flood (Fl), pollution (Po), siltation (St), and water turbidity (Tu). Changes: diminished fishing activities (FD), loss of crops (CL), shortages of food and water (WS), effects on tourism (TI), housing damage (HD), health problems (HA), and disruptions to the supply of electricity (E).

cipants. This may be because perceptions do not only depend on physical characteristics but also on the properties that each person attributes to any environmental component as a result of their knowledge and experience; hence, it is important to identify the perceptions of people to understand the motivations that determine their decision-making (Marín-Muñíz et al. 2016). This may help to explain why even users engage in activities (e.g., overfishing, cutting mangroves, and dumping garbage) that go against their own understanding of the importance of wetlands.

Wetland abundance can also be a major factor that influences how people value these ecosystems. This hypothesis was supported in this study. Even though saltmarsh ES are largely unknown and underestimated (Lebreton et al. 2019) and although there are few collective benefits to shrimp farming, these ecosystems were classified as priority wetlands in the three communities of Sinaloa, in which the largest wetland expanse in Mexico is located. In contrast, these two wetland types were not considered to be of priority in the community of BCS where these two wetland types are scarce (Berlanga-Robles et al. 2011; 2021). It is important to mention that the LR workshop participants held different occupations to those of the participants of the three communities in Sinaloa and were mainly employees of civil society organizations and the government. As such, these individuals were not direct users of the wetlands like the participants of the three workshops in Sinaloa. It is perhaps not surprising that mangroves were well valued in LR despite their minor presence in BCS (Velázquez-Salazar et al. 2021).

At the national level, mangroves, their associated benefits, and the threats to these ecosystems are some of the most important issues at all levels of the environmental debate within Mexico. Multiple important contributions to understanding this ecosystem type in the Gulf of California (e.g., their extensions, contributions to fishing productivity, and economic valuations) have been produced by researchers (e.g., Aburto-Oropeza et al. 2008, Ruiz-Luna et al. 2010, Hernández-Guzmán et al. 2021) and governmental institutions (https://www.biodiversidad.gob.mx/monitoreo/smmm/extensionDist).

Maltby and Acreman (2011) and Marín-Muñíz et al. (2016) agree that wetland loss and degradation are due in part to a poor understanding of how wetlands function and the roles they play in providing ES to people living within or near them. When considering provisioning and maintenance ES, the results obtained in this study suggest that the participants clearly understood the direct benefits or provisioning services of the wetland types in this study. In contrast, regulating and supporting and cultural services were less recognized. Thus, there is a need to educate the public on the importance of wetlands at the local level. This result may help guide and improve both existing and future environmental education, management, decision-making, and conservation actions. Addressing this need is especially important given that many people do not yet fully recognize how wetland ES help communities mitigate and adapt to the effects of climate change.

The knowledge of the participants in this study of ES is likely connected to their experiences and interactions with neighboring wetlands. Therefore, fishing, food provisioning, and the maintenance of biodiversity (e.g., habitats and refuge areas for various species including those of commercial interest) are the most easily identifiable ES in communities orientated towards fishing and aquaculture. It is important to point out that the wetlands adjacent to the communities included in this study are focus areas of various civil society organizations given their national or international status as protected areas (e.g., Ramsar sites). As such, various conservation activities are conducted in these sites, which may serve to reinforce the understanding of community members of the importance of wetlands and their ES. The perceived threats to wetlands were also similar among the communities in this study, although differences were present between LR in BCS and the three communities of Sinaloa. The threats that were perceived in all communities included poor fishing practices or water contamination, which have been previously documented, and pollution caused by various agricultural sources and untreated urban wastewater discharge (Arellano-Aguilar 2017; Moeder et al. 2017).

It is important to point out that all communities in this study were aware of environmental changes, which were mainly centered around the landscape and climate. Both of these types of change have various repercussions on socioeconomic activities and the environment that vary in intensity (e.g., floods, pollution, sedimentation, water turbidity, reductions in fishing activities, poor harvests, food shortages, and damage to housing). Therefore, preventative actions are required to adapt to future change and thus reduce threats to human health and well-being. These actions should consider regional differences among local livelihoods, resource availability, and consumption patterns (McMichael 2013).

If the flow and supply of ES depends on the integrity of ecosystems and their resilience, then it is critical to implement efficient management measures to conserve and restore wetlands to ensure they retain their functionality even after extreme meteorological phenomena like ERE. For this, resource users must participate in decision-making processes, which must be based on the best available information. In this sense, ES must be recognized and valued to further conservation, resource management, and environmental planning efforts that concurrently generate ecological, social, and economic benefits (De Groot et al. 2010). Social participation has become an internationally recommended component of conservation programs. For example, international organizations such as the United Nations Environment Program, have established strategies to promote the active participation of local communities to ensure environmental protection (Artigas et al. 2014). In addition, Agenda 21 of the United Nations recommends adopting an integrated and participatory approach to conservation and development (Agenda 21 1992). In addition, analyses of attitudes, beliefs, and sociocultural preferences that societies have of ES can be relevant to decision-making and environmental planning processes, as the success of these policies heavily depends on social acceptance and support (García-Llorente et al. 2020). Nevertheless, international recommendations of social participation in decision-making have not yet been embraced in Latin America, in which participation is limited to priority problems in protected areas (Maldonado et al. 2020).

Information on the perceptions and importance of wetlands in Mexico is scarce. To a large extent, this information has been generated as a result of social conflicts due to population growth and territorial expansion, which have traditionally endangered the existence of wetlands and other ecosystems. Therefore, the results of this study serve to fill this information gap and may be used to provide guidelines to structure education, conservation, and resource management strategies in the study region, which may be applied to other communities in different regions. The results of this study also serve to highlight the importance of wetlands and the ES they offer, regardless of their locations or characteristics. Wetlands can be underestimated by societies and communities that remain unaware of their importance. For example, at the regional level in Mexico, the expanse of wetlands in Sinaloa was reduced by  $\sim 580 \ km^2$  from 1995 to 2019 (Hernández-Guzmán et al. 2021). By

educating people at local, regional, and national levels on the importance of wetlands, losses like the one in Sinaloa can be reversed.

# CONCLUSION

To conclude, wetland conservation and the wise use of wetland resources requires both technical solutions and the inclusion of resource users in the decisionmaking process. By initiating and encouraging the participation of resource users in management actions and ensuring continuity to participatory processes, important traditional knowledge can be included in conservation strategies. In addition, community participation and collaboration with management authorities can lead to the discovery of beneficial alternatives to established actions and novel solutions to problems that affect the livelihoods and wellbeing of resource users. These actions can then be scaled from the community level to develop regional management and resource utilization strategies.

# ACKNOWLEDGMENT

The authors greatly appreciate the economic support by the Consejo Nacional de Ciencia y Tecnología – Mexico (CONACYT project number PN-2017-01-4764). We also acknowledge the support by The David and Lucille Packard Foundation (Grant 2022-73257). This work was possible thanks to the support and collaboration of the inhabitants of the localities where the community workshops were held (Isla del Bosque, Cristo Rey, Las Arenitas and Loreto). Thanks to all of them.

# DATA AVAILABILITY

Data collected in this research is available upon reasonable request to the corresponding author.

# CONFLICT OF INTEREST

The author has no conflicts of interest to declare.

# CONTRIBUTION STATEMENT

Conceived of the presented idea: ACE, ARL. Investigation: ACE, ARL, CABR. Carried out the data analysis: CABR. Wrote the first draft of the manuscript: ACE, ARL. Review and final write of the manuscript: ACE, ARL, CABR.

## REFERENCES

Aburto-Oropeza O, Ezcurra E, Dannemann G, Sala E (2008) Mangroves in the Gulf of California increase fishery yields. *PNAS* doi: 10.1073/p-nas.0804601105.

Agenda 21 (1992) https://sustainabledevelopment. un.org/outcomedocuments/agenda21. Accessed 9 September 2022.

Agresti A (2007) An Introduction to Categorical Data Analysis. 2 ed. Wiley, Canada.

Arellano-Aguilar O, Betancourt-Lozano M, Aguilar-Zárate G, Ponce de Leon-Hill C (2017) Agrochemical loading in drains and rivers and its connection with pollution in coastal lagoons of the Mexican Pacific. *Environmental Monitoring and Assessment* doi: 10.1007/s10661-017-5981-8.

Artigas PE, Ramos RAE, Vargas RH (2014) La participación comunitaria en la conservación del medioambiente: Clave para el desarrollo local sostenible. *DELOS* 7(21).

Berlanga-Robles CA, Ruiz-Luna A, de la Lanza G (2008) **Esquema de clasificación de los humedales en México.** Investigaciones Geográficas, Boletín del Instituto de Geografía, UNAM. 66:25-46.

Berlanga-Robles CA, Ruiz-Luna A, Bocco G, Vekerdy Z (2011) Spatial analysis of the impact of shrimp culture on the coastal wetlands on the Northern coast of Sinaloa, Mexico. Ocean and Coastal Management 54:535-543.

Berlanga-Robles CA, Ruiz-Luna A, Hernández-Guzmán R (2021) Caracterización socioeconómica y ambiental de la zona costera del noroeste de México, por medio de indicadores. In: Cervantes-Escobar A, Hernández-Guzmán R, Montijo-Galindo A (eds) Vulnerabilidad costera en el noroeste de México. Un enfoque multidisciplinario. CIAD, Mazatlán, México, pp. 15-42.

Camacho-Valdez V, Ruiz-Luna A (2012) Marco conceptual y clasificación de los servicios ecosistémicos. *Revista Bio Ciencias* doi: 10.15741/revbio.01.04.02.

Camacho-Valdez V, Ruiz-Luna A, Ghermandi A, Nunes PALD (2013) Valuation of ecosystem services provided by coastal wetlands in northwest Mexico. Ocean and Coastal Management doi: 10.1016/j.ocecoaman.2013.02.017.

Camacho-Valdez V, Ruiz-Luna A, Ghermandi A, Berlanga-Robles CA, Nunes PALD (2014) Effects of land use changes on the ecosystem service values of coastal wetlands. *Environmental Manage*-

ment doi: 10.1007/s00267-014-0332-9.

Carpenter SR, Fisher SG, Grimm NB, Kitchell JF (1992) Global change and freshwater ecosystems. Annual Review of Ecology and Systematics 23:119-139.

Carrera E, de la Fuente G (2003) **Inventario y clasificación de humedales en México.** Parte I. Ducks Unlimited de México, A.C., México.

Carvajal AG (1994) Clima, suelo, boque y sus interrelaciones en la percepción de los Talamaqueños, Costa Rica. Anuario de Estudios Centroamericanos 20(1):43-64.

Clarkson BR, Ausseil AE, Gerbeaux P (2013) Wetland ecosystem services. In: Dymond JR (ed) Ecosystem services in New Zealand – conditions and trends. Manaaki Whenua Press, Lincoln, New Zealand.

CONABIO (Comisión Nacional para el Conocimiento y Uso de la Biodiversidad). http://www.conabio. gob.mx/conocimiento/regionalizacion/doctos/ terrestres.html Accessed 14 September 2022.

CONAGUA (Comisión Nacional del Agua) https: //www.gob.mx/conagua/acciones-y-programas/ inventario-nacional-de-humedales-inh Accessed 20 January 2023.

CONEVAL (2021) https://municipal-coneval. hub.arcgis.com Accessed: 9 September 2022.

Costanza R, De Groot R, Braat L, Kubiszewski I, Fioramonti L, Sutton P, Farber St, Grasso M (2017) Twenty years of ecosystem services: How far have we come and how far do we still need to go? *Ecosystem Services* doi: 10.1016/j.ecoser.2017.09.008.

Data Mexico (2021) https://datamexico.org/es/ profile/geo/loreto?redirect=true Accessed 9 September 2022.

De Groot RS, Alkemade R, Braat L, Hein L, Willemen L (2010) Challenges in integrating the concept of ecosystem services and values in landscape planning, management and decision. *Ecological Complexity* doi: 10.1016/j.ecocom.2009.10.006.

Duarte CM, Alonso S, Benito G, Dachs J, Montes C, Pardo M, Ríos AF, Simó R, Valladares F (2006) Cambio global. Impacto de la actividad humana sobre el sistema Tierra. España: Consejo Superior de Investigaciones Científicas.

Émerit B (2008) Cambio climático y sustentabilidad económica y social: Implicaciones sobre el bienestar social CIRIEC-ESPAÑA. *Revista de Economía Pública, Social y Cooperativa* 61:51-72. Flores VFJ, Casasola P, de la Lanza-Espino G, Agraz HC (2010) **El manglar, otros humedales costeros y el cambio climático.** In: Botello AV, Villanueva-Fragoso S, Gutiérrez J, Rojas GJL (ed.). Vulnerabilidad de las zonas costeras mexicanas ante el cambio climático. Semarnat-Ine, Unam-Icmyl, Universidad Autónoma de Campeche, pp.165-188.

García-Llorente M, Castro AJ, Quintas-Soriano C, Oteros-Rozas E, Iniesta-Arandia I, González JA, García del Amo D, Hernández-Arroyo M, Casado-Arzuaga I, Palomo I, Gómez-Baggethun E, Onaindia M, Montes C, Martín-López B (2020) Local perceptions of ecosystem services across multiple ecosystem types in Spain. Land doi: 10.3390/land9090330.

Habershon N (1993) Metaplan (R): Achieving two-way communications. Journal of European Industrial Training doi: 10.1108/03090599310042528.

Hernández-Guzmán R, Berlanga-Robles CA, Ruiz-Luna A, Suazo-Ortuño I (2021) Cambios en la extensión de los humedales costeros del estado de Sinaloa, México. In: Cervantes-Escobar A, Hernández-Guzmán R, Montijo-Galindo A (eds.). Vulnerabilidad costera en el noroeste de México. Un enfoque multidisciplinario. 1 ed. CIAD, pp. 147-164..

Lebreton B, Rivaudb A, Picota L, Prévostb B, Barilléc L, Sauzeaud T, Beseres Pollacke J, Lavauda J (2019) From ecological relevance of the ecosystem services concept to its sociopolitical use. The case study of intertidal bare mudflats in the Marennes-Oléron Bay, France. Ocean and Coastal Management doi: 10.1016/j.ocecoaman.2019.01.024.

MacDonald PL, Gardner RC (2000) Type I error rate comparisons of post hoc procedures for  $I \times J$  chi-square tables. Educational and Psychological Measurement doi: 10.1177/00131640021970871.

Maldonado IOA, Chávez DRM, Bravo OML (2020) Áreas naturales protegidas y participación social en América Latina: problemas y estrategias para lograr la integración comunitaria. *Región y Sociedad* doi: 10.22198/rys2020/32/1277.

Maltby E, Acreman MC (2011) Ecosystem services of wetlands: pathfinder for a new paradigm. *Hydrological Sciences Journal* doi: .

Mangiafico SS (2015) An R Companion for the Handbook of Biological Statistics, version 1.3.4. https://rcompanion.org/documents/ RCompanionBioStatistics.pdf.

Marín-Muñiz JL, Hernández AME, Silva RE, Moreno CP (2016) Percepciones sobre servicios ambientales y pérdida de humedales arbóreos en la comunidad de Monte Gordo, Veracruz. *Madera y Bosques* 22:53-69.

McDonald JH (2014) Handbook of Biological Statistics. Sparky House Publishing, Baltimore, Maryland. http://www.biostathandbook. com/#print.

McMichael AJ (2013) Globalization, climate change, and human health. *New England Journal of Medicine*. doi: 10.1056/NEJMra1109341.

Millennium Ecosystem Assessment (2005) Ecosystem and human well-being: wetlands and water synthesis. World Resources Institute, Washington, DC. https://wedocs.unep.org/20. 500.11822/8735.

Mitsch WJ (2016) Wetlands and climate change. National Wetlands Newsletter 38(1): 5-11. https:// www.researchgate.net/publication/291586979\_ Wetlands\_and\_Climate\_Change.

Mitsch WJ, Gosselink JG (2015) **Wetlands.** 5ed. Wiley and Sons, Inc.

Moeder M, Carranza-Diaz O, López-Angulo G, Vega-Aviña R, Chávez-Durán FA, Jomaa S, Delgado-Vargas F (2017) Potential of vegetated ditches to manage organic pollutants derived from agricultural runoff and domestic sewage: A case study in Sinaloa (Mexico). The Science of the Total Environment doi: 10.1016/j.scitotenv.2017.04.149.

Olmsted I (1993) Wetlands of Mexico. In: Whigham DF, Dykyjová D, Hejný S. (eds) Wetlands of the world: Inventory, ecology and management Volume I. Handbook of vegetation science, 15(2). Springer, Dordrecht, pp. 637–678.

Ottinger M, Clauss K, Kuenzer C (2016) Aquaculture: Relevance, distribution, impacts and spatial assessments–A review. Ocean and Coastal Management. doi: 10.1016/j.ocecoaman.2015.10.015.

Ramsar Convention (2008) **Resolution X.3 The Changwon Declaration on human well-being and wetlands**. Changwon, Republic of Korea. https://medwet.org/wp-content/uploads/2012/ 10/COP10\_key\_res\_x\_03\_Changwon\_Declaration\_ 2008.pdf.

Rojas O, Zamorano M, Saez K, Rojas C, Vega C, Arriagada L, Basnou C (2017) Social perception of ecosystem services in coastal wetland postearthquake: A case study in Chile. *Sustainability.* doi: 10.3390/su9111983. Ruiz-Luna A, Acosta VJ, Berlanga-Robles AC (2008) On the reliability of the data of extent of mangrove; a case study in Mexico. Ocean and Coastal Management 51:342-351.

Ruiz-Luna A, Cervantes-Escobar A, Berlanga-Robles C (2010) Assessing Distribution Patterns, Extent, and Current Condition of Northwest Mexico Mangroves. *Wetlands* 30:717-723.

Sage RF (2020) Global change biology: A primer. Global Change Biology doi: 10.1111/gcb.14893.

SEMARNAT (2012) Los humedales en México. Oportunidades para la sociedad. https://biblioteca.semarnat.gob.mx/janium/ Documentos/Cecadesu/Libros/Humedales.pdf.

Stern N (2007) **The economics of climate change: The Stern review.** Cambridge Univ. Press. doi: 10.1017/CBO9780511817434.

TEEB (2013) The economics of ecosystems and biodiversity for water and wetlands. London and Brussels, Institute for European Environmental Policy (IEEP) & Ramsar Secretariat..

Velázquez-Salazar S, Rodríguez-Zúñiga MT, Alcántara-Maya JA, Villeda-Chávez E, Valderrama-Landeros L, Troche-Souza C, Vázquez-Balderas B, Pérez-Espinosa I, Cruz-López MI, Ressl R, De la Borbolla DVG, Paz O, Aguilar-Sierra V, Hruby FC, Muñoa-Coutiño JH (2021) Manglares de México. Actualización y análisis de los datos 2020. CONABIO. México. [https://agua.org.mx/biblioteca/manglaresde-mexico-actualizacion-y-analisis-de-losdatos-2020/].

Vernon ME, Campos BR, Burnett RD (2019) A guide to climate-smart meadow restoration in the Sierra Nevada and southern Cascades. *Point Blue Contribution*, (2232). https:// data.pointblue.org/apps/snamin/uploads/2019% 20Climate%20Smart%20Meadow%20Restoration\_ MarianE.Vernon.pdf.

West LJ, Hankin RK (2008) Exact tests for twoway contingency tables with structural zeros. *Journal of Statistical Software* 28:1-19.

> Received: 11 March 2022 Accepted: 07 February 2023 Published: 16 February 2023

Editor: Norma Hilgert